Author's response to reviews

Title: Diagnosis of tuberculosis in an Indian population by an indirect ELISA protocol based on detection of Antigen 85 complex: a prospective cohort study

Authors:
- Rajpal S. Kashyap (raj_ciims@rediffmail.com)
- Anju N Rajan (anju_rajann@rediffmail.com)
- Sonali S Ramteke (sonaliramteke@yahoo.co.in)
- Vijay S. Agrawal (ciims_ngp@sancharnet.in)
- Sanjivani S Kelkar (ciims_ngp@sancharnet.in)
- Hemant J. Purohit (hemantdrd@hotmail.com)
- Girdhar M. Taori (taorigm_ciims@yahoo.co.in)
- Hatim F. Daginawala (hfd_ciims@rediffmail.com)

Version: 2 Date: 28 December 2006

Author's response to reviews:

MS: 3516515591255519
Diagnosis of tuberculosis in Indian population with an Indirect ELISA protocol based on Antigen 85 complex detection assay: A prospective cohort study.
Rajpal S. Kashyap, Anju N Rajan, Sonali S Ramteke, Vijay S. Agrawal, Sanjivani S Kelkar, Hemant J. Purohit, Girdhar M. Taori and Hatim F. Daginawala
BMC Infectious Diseases

To,
The Editor,
BMC Infectious Diseases.

Sir,

Please find enclosed the revised manuscript (above).

As per your suggestion for improving the style of written English in the manuscript, we have revised the manuscript with the help Manuscript Preparation Service (Biomedes). The changes made in the manuscript are indicated with blue color.

Samples were collected from all study groups for which participants consent was obtained

Sincerely

Dr. H.F.Daginawala
Research Lab
Central India Institute Of Medical Sciences
Nagpur-10.
India

Dated: 28/12/06